Re: 'Endothelial Nitric Oxide Synthase Gene Polymorphism (Glu298Asp) and Development of Pre-eclampsia: a nested case-control study and a meta-analysis.'

Manuscript: 1452598236642891; Version 2.

Dear Editor,

Thank you for the comments of the referee that have been used to prepare the enclosed revised manuscript. We now provide the data as requested by the reviewer:

Thank you

Christina Yu
Harris Birthright Research Centre for Fetal Medicine,
King’s College Hospital,
Denmark Hill,
London SE5 9RS, U.K.
Tel: 0044 207 346 3070
Email chrissieyu@aol.com

Juan P. Casas
Department of Epidemiology and Population Health
London School of Hygiene & Tropical Medicine
Email: Juan.pablo-Casas@lshtm.ac.uk
Reviewer 2

1. My first issue is their reluctance to provide the non-significant estimates for the covariates tested arguing that "by doing that, we may disperse the readers attention from the genetic factors". In genetic epidemiology, a field abundant in spurious findings due to multiple testing and chance (both in the tested association itself as in the environmental interactions), reporting non-significant findings is encouraged. BioMed central, being an online journal with no word or table limitations, offers an excellent opportunity for investigators to present their results properly. The author's argument is therefore not convincing.

Response: The estimates for the non-genetic variables included in the main multivariate model are now reported in the manuscript, as a new table 3.

2. The authors did not respond appropriately to my comment about performing a stratified analysis (for example by ethnicity) for the case control study. My request is to provide ORs for each ethnic group. The approach that the authors have adopted (to report and test the ethnic distribution in cases and controls) is not informative enough.

Response: The genetic estimates stratified by ethnic group are now reported in the table 2.